Author's response to reviews

Title: Acute dengue virus 2 infection in Gabonese patients is associated with an early innate immune response, including strong interferon alpha production

Authors:

Pierre Becquart (pierre.becquart@ird.fr)
Nadia Wauquier (wauquiernadia@orange.fr)
Dieudonné Nkoghe (dnkoghe@hotmail.com)
Angélique Ndjoyi-Mbiguino (ainanga@yahoo.fr)
Cindy Padilla (padilla.cindy@gmail.com)
Marc Souris (marc.souris@ird.fr)
Eric M Leroy (eric.leroy@ird.fr)

Version: 7 Date: 13 December 2010

Author's response to reviews: see over
To the Editor

BMC Infectious Diseases

Dear Sir,

Please find enclosed our amended manuscript entitled *Acute dengue virus 2 infection in Gabonese patients is associated with an early innate immune response, including strong interferon alpha production.*

**Major revisions**

*Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is ‘First name Initial Last name’ (e.g. Joe F Bloggs)*

Author list corrected as requested in the submission system (Eric M LEROY).

**Minor revisions**

*Email address: please include email addresses for all the authors on the title page, using the following format:*

AB: abcd@institution.ac.uk
EF: efg@generic.co.uk
IJ: IJKL@corporation.com

Email addresses corrected as requested